# UC Irvine

UC Irvine Previously Published Works

## Title

Associations Between Rurality, pre-pregnancy Health Status, and Macrosomia in American Indian/Alaska Native Populations

Permalink https://escholarship.org/uc/item/0vk472m0

Journal Maternal and Child Health Journal, 26(12)

ISSN

1092-7875

Authors

Hiratsuka, Vanessa Y Reid, Margaret Chang, Jenny <u>et al.</u>

Publication Date 2022-12-01

DOI 10.1007/s10995-022-03536-w

Peer reviewed

# Associations Between Rurality, pre-pregnancy Health Status, and Macrosomia in American Indian/Alaska Native Populations

Vanessa Y. Hiratsuka<sup>1,2</sup>, Margaret Reid<sup>3</sup>; Jenny Chang<sup>4</sup>; Luohua Jiang<sup>5</sup>; Angela G. Brega<sup>6</sup>; Amber L. Fyfe-Johnson<sup>7</sup>; Kimberly R. Huyser<sup>8</sup>; Michelle Johnson-Jennings<sup>9</sup>; Cheryl Conway<sup>10</sup>; John F. Steiner<sup>11</sup>; Jennifer Rockell<sup>12</sup>; Denise A. Dillard<sup>1</sup>; Kelly Moore<sup>13</sup>; Spero M. Manson<sup>13</sup>; Joan O'Connell<sup>14</sup>

## Abstract

**Objectives** To examine the relationships between pre-pregnancy diabetes mellitus (DM), gestational diabetes mellitus (GDM), pre-pregnancy body mass index (BMI) and county-level social determinants of health, with infant macrosomia within a sample of American Indian/Alaska Native (AI/AN) women receiving Indian Health Service (IHS) care.

**Methods** The sample included women-infant dyads representing 1,136 singleton births from fiscal year 2011 (10/1/2019-9/30/2011). Data stemmed from the IHS Improving Health Care Delivery Data Project. Multivariate generalized linear mixed models were fitted to assess the association of macrosomia with pre-pregnancy health status and social determinants of health.

**Results** Nearly half of the women in the sample were under age 25 years (48.6%), and most had Medicaid health insurance coverage (76.7%). Of those with a pre-pregnancy BMI measure, 66.2% were overweight or obese. Although few women had pre-pregnancy DM (4.0%), GDM was present in 12.8% of women. Most women had a normal term delivery (85.4%). Overweight, obesity, pre-pregnancy DM, and county-level rurality were all significantly associated with higher odds of infant macrosomia.

# **Significance Statement**

Birthweight is associated with the risk of metabolic syndrome, diabetes, and obesity in adulthood. Previous studies within American Indian and Alaska Native (AI/AN) populations on macrosomia have not included social determinants of health risk factors. Early life environments experienced by women, such as disadvantaged neighborhood, built environment factors and community-level socioeconomic factors are related to high birthweight and high pre-pregnancy weight. AI/AN people experience persistent health disparities across the lifespan compared to non-Hispanic whites and experience greater prevalence of poor health outcomes. Knowledge is lacking on infant outcomes as a function of maternal factors in the AI/AN population. The current study found that AI/AN women who were overweight or obese, had pre-pregnancy diabetes, or lived in a rural county had higher odds of having an infant with a high birthweight. AI/AN women who may become pregnant should be advised on weight gain and diabetes to help decrease the risk of high birthweight and later childhood obesity.

### Introduction

Early life weight is associated with the risk of metabolic syndrome, diabetes, and obesity in adulthood (Dixon et al., 2012; Marshall, 2019; Pachucki & Goodman, 2015). Systematic studies and meta-analyses have shown that birth weight, an indicator of early life development, is associated with impaired glucose tolerance, insulin resistance, and cardiovascular disease later in the life course (Knop et al., 2018; Tian et al., 2019). Childhood health outcomes result from exposures during the prenatal period, as women who are overweight, obese, or who have prepregnancy diabetes mellitus (DM) or gestational diabetes mellitus (GDM) are more likely to have a newborn with macrosomia (infant birth weight > 4000 g) (Garcia-Vargas et al., 2012; Halvorson et al., 2017; Hunt et al., 2013; Voerman et al., 2019). Infants with macrosomia are more likely to be overweight or obese in childhood and adulthood, and to have metabolic syndrome or diabetes mellitus (DM) later in life (Boney et al., 2005; Hu et al., 2020; Knop et al., 2018; Tian et al., 2019; Yu et al., 2011). Early life environments experienced by women, such as disadvantaged neighborhood and built environment factors (e.g., rurality, limited access to grocery stores) and community-level socioeconomic factors (e.g., limited education, high poverty), are related to infant macrosomia (Carter et al., 2012; Shi et al., 2013; Vaiserman, 2017).

Social determinates of health (SDOH) are occurrences that are influenced by the political, social, environmental, and economic conditions in which people are "born, grow, live, work, age," and receive health care (Link & Phelan, 1995; Mitchell, 2012). The relationship between adverse maternal social and environmental influences and high birth weight has been an area of concern by the Indian Health Service (IHS) since the 1960's (Rosa & Resnick, 1965; Thierry et al., 2009). American Indian/Alaska Native (AI/AN)s experience persistent health disparities across the lifespan compared to non-Hispanic whites and experience greater prevalence of poor health outcomes (e.g., diabetes, overweight/obesity, cardiovascular diseases) (Chang et al., 2016; Jones, 2006; Lanier, 2007). The drivers of health disparities are complex and persevere across generations with interactions between multiple SDOH, such as household income, educational attainment, and food insecurity, that influence poor health outcomes (Dagher & Linares, 2022; Marmot et al., 2008). The root determinates of disruptions to AI/AN health include colonization, medical pathologizing, social exclusion, racial discrimination, oppression, marginalization, and chronic inadequacy of funding of the IHS (Christensen & Damon, 2022; Jaramillo et al., 2022; Kruse et al., 2022).

AI/AN children have a higher prevalence of obesity than US children overall (Bullock et al., 2017; Ogden et al., 2015; Schell & Gallo, 2012). The prevalence of pediatric obesity has rapidly increased in the US over the last 30 years, with notable disparities in racial and ethnic minority populations (Ogden et al., 2015). AI/AN pediatric obesity is associated with obesogenic influences during gestation, which predict offspring obesity later in the life course (Dixon et al., 2012; Lynch & Smith, 2005). Women's risk factors for AI/AN pediatric obesity include pre-pregnancy DM, prepregnancy overweight or obesity, smoking while pregnant, late onset of prenatal care, high gestational weight, gestational weight gain, GDM, and maternal depression (Schell & Gallo, 2012). SDOH risk factors for AI/AN pediatric obesity include high levels of poverty, unemployment, and food insecurity, and low levels of education in the community

(Schell & Gallo, 2012). Few studies have examined infant outcomes as a function of maternal factors in the AI/AN population (Anderson et al., 2016; LaVallie et al., 2003; Rockhill et al., 2015; Schell & Gallo, 2012). AI/AN children of women with pre-pregnancy DM or GDM are nearly two times more likely to be obese and 1.71 times more likely to be overweight by ages 5–8 years than are children of women without pre-pregnancy DM or GDM. Within AI/AN populations, overweight, obesity, and preexisting DM are all associated with increased odds of macrosomia (Anderson et al., 2016; Rockhill et al., 2015).

Past studies in the AI/AN population have not assessed the role of community-level SDOH risk factors in addition to maternal risk factors (e.g., diabetes, obesity). In this study, we examined the association of maternal risk factors and SDOH with macrosomia among AI/AN families accessing services through IHS. IHS clinics and programs are located on or near AI reservations and AN communities, locations that are predominantly rural and impoverished (Brenneman et al., 2006). We hypothesize that mothers' obesity, prepregnancy DM, are significantly associated with higher odds of infant macrosomia; meanwhile, lower level of neighborhood socioeconomic status (i.e., lower income, education, and rurality) are also significantly associated with higher odds of infant macrosomia. The study data were extracted from a data infrastructure developed for the IHS Improving Health Care Delivery Data Project (IHS Data Project). Although the IHS Data Project was not originally created to examine AI/AN women's risks for high birthweight infants, we were able to use data from a subset of project sites to examine macrosomia among infants of AI/AN women.

#### Methods

#### **Data Source**

Since 2010, the Centers for American Indian and Alaska Native Health at the Colorado School of Public Health, has collaborated with the IHS to create a longitudinal data infrastructure that houses health status, service utilization, and treatment cost data for over 640,000 AI/ANs who lived throughout the United States, representing nearly 30% of AI/ANs who use IHS services (O'Connell et al., 2014). The data infrastructure is a synthesis of existing electronic data from multiple IHS platforms and includes data for seven fiscal years (FY) 2007–2013. FY 2011, for example, is October 2010 to September 2011. The IHS Data Project provides a purposeful sample of AI/AN persons who lived in one of 15 IHS Service Units (hereafter referred to as project sites) located throughout the United States. The IHS Data Project includes one project site in the East, four in the Northern Plains, two in the Southern Plains, five in the Southwest, two in the Pacific Coast, and one in Alaska, based on geographic areas reported elsewhere (O'Connell et al., 2014). The HIS Data Project population was identified by geographic area (i.e., Service Units), rather than by random sampling, to create important community and county measures not available elsewhere (O'Connell et al., 2014). The IHS Data Project population is comparable to the national IHS service population in terms of age and gender (Indian Health Service, 2010). More information about the data infrastructure may be found elsewhere (O'Connell et al., 2014).

#### **Study Sample**

Women were identified using obstetric admissions using the inpatient service code for 'obstetric' visits during FY2007 - FY2013. Of the fifteen project sites, four had no or almost no obstetric admissions and one site was excluded due to the large percentage of women referred to this site from other locations who may have had more complicated pregnancies. Next, we limited the obstetric admissions to individuals with a labor and delivery ICD-9 diagnostic code. If a mother had multiple admissions within 9 months, the first admission was kept as the delivery admission. We then identified infants from inpatient data using the inpatient service code for 'newborn', limiting admissions to instances where the month and year of service matched the date in the demographic information.

Mothers and their infants were linked using the delivery hospital, dates of admission, and either the household identification number or the community of residence, both from the IHS registration data. We limited this analysis to singleton births. Multiple births are excluded as they are more likely to be of a low birthweight or premature. Among the ten remaining project sites, only six sites had 75% or more of deliveries linked to newborns. Thus, we limited the study sample to those six sites.

#### **Study Variables**

*Primary Outcome.* Infant weight, the primary outcome, was measured in grams at birth. Macrosomia was defined as a weight  $\geq$  4000 g regardless of gestational age (American College of Obstetricians and Gynecologists, 2020; Duryea et al., 2014).

Women's demographic and health coverage data. Women's month and year of birth were used to calculate the women's age at the service date when the child was born and categorized into age groups. Health care insurance coverage was categorized into three categories: Medicaid, private insurance, or no coverage in addition to access to IHS services.

Women's health status. Inpatient and outpatient data provided ICD-9 codes indicating GDM (648.80 to 648.84). Prepregnancy DM was determined by using data on ICD-9-CM diagnoses, procedure codes, medication use, and blood glucose control (i.e., hemoglobin A1c  $\geq$ 6.5%) in the two years prior to the pregnancy. A woman's pre-pregnancy body mass index (BMI) was defined as the BMI measure that was closest to the gestation date (defined as 270 days prior to the delivery date). BMI measures that were more than one year before the gestation date, more than one month after the gestation date, or during a prior pregnancy were excluded. We excluded biologically implausible height, weight, and BMI values for adults aged 18 years and older. We used the standard definitions for obesity for individuals age 18 and older (overweight: BMI = 25.0-30.0 kg/m2; obese: BMI > 30.0 kg/m2). Relevant chronic conditions of women were identified using ICD-9 codes reported in the inpatient and outpatient service utilization records in FY2010, supplemented by medication data. Sightlines<sup>™</sup> DxCG Risk Solutions software groups ICD-9 codes into diagnostic cost groups (Verisk Health, 2011). Diagnostic cost groups are employed by the federal government and private insurers to identify chronic conditions. We used this software to identify adults with one or more types of chronic conditions: hypertension, cardiovascular disease, mental health disorder, and substance use disorder (inclusive of drug, alcohol, and tobacco use). Pre- and post-term delivery were identified by ICD-9 diagnosis codes.

SDOH indicators. SDOH indicators were largely county level measures drawn from Census Bureau and United States Department of Agriculture (USDA) Food Environment Atlas data sources and were added to the IHS Data Project by county of residence (Indian Health Service, 2010; US Census Bureau, 2001, 2015, 2016). From the Census, we had county-level indicators from the 2010–2014 American Community Survey (ACS) and decennial Census 2000 and 2010 (US Census Bureau, 2001, 2015). We used the Healthy People 2020 SDOH framework to organize and categorize our SDOH independent variables. For each SDOH, we calculated the median value for the 72 counties included in our data and created a binary variable of SDOH based on the median value of it (i.e.,  $\leq$  median vs. > median).

Education and economic stability measures were drawn from 2010 to 2014 ACS countylevel data. Estimates were calculated for AI/AN persons who report accessing HIS services in the ACS response (US Census Bureau, 2016). Educational attainment was calculated as the percentage of adults aged 25 years and older who did not complete high school. We defined the percentage of households with a low income as the percentage of households with incomes below 139% of the federal poverty level (US Census Bureau, 2001, 2015, 2016). Neighborhood and built environment measures were from the USDA Food Environment Atlas, including the percent of the county population with low access to a grocery store in 2010 and the number of Supplemental Nutrition Assistance Program (SNAP) authorized stores per 1,000 population (Indian Health Service, 2010). We also coded county rurality using the National Center for Health Statistics urban-rural classification (National Center for Health Statistics, 2017).

#### **Statistical Analysis**

Descriptive statistics of the woman's individual demographic and health status characteristics and county-level SDOH were examined for the full study sample and stratified by infant macrosomia status. Multivariate generalized linear mixed models were fitted to assess the association of high birthweight with maternal health factors (i.e., DM, GDM, and obesity) and SDOH indicators, with county-level random intercepts to account for clustering of observations within counties. We subsequently introduced potential confounding covariates in steps (Fig. 1). Model 1 included maternal sociodemographic and health factors (i.e., age, health insurance, pre-pregnancy BMI category, pre-pregnancy DM, and GDM status) and pre-/post-term birth outcome. Model 2 added comorbidities (i.e., hypertension, cardiovascular disease, mental health disorder, substance use disorder) to Model (1) Model 3 added county-level neighborhood and built environment SDOH variables to Model (2) Model 4 added county-level education and economic stability variables to Model (3) Adjusted odds ratios and their p values are presented. Statistical significance was defined as p < 0.05 with 95% confidence intervals. We used SAS® 9.4 and Stata statistical software to conduct descriptive and multivariable analyses (SAS Institute, 2013; Stata-Corp, 2015).

#### **Institutional and Tribal Approval**

Project personnel work in a collaborative manner with HIS and the Tribal organizations that participate in the IHS Data Project. This collaboration takes place through the project's Collaborative Network, which includes three advisory committees (i.e., Steering, Project Site, and Patient) and a process to obtain approvals from the IHS Institutional Review Board (IRB), Tribal IRBs, and Tribal Councils and Authorities, in addition to the university's IRB (Colorado

Multiple Institutional Review Board). Relevant IRB and community level approval of all aspects of this study were received prior to the conduct of the investigation. Patients at study locations gave their informed consent for deidentified patient data to be used for health services research.

#### Results

Table 1 presents descriptive characteristics for women stratified by infant birth weight. Nearly half of women in the sample were under age 25 years (48.6%) and most had Medicaid health insurance coverage (76.7%). Of those with a pre-pregnancy BMI measure, 66.2% were overweight or obese. Few had a pre-pregnancy diagnosis of DM (4.0%), while 12.8% experienced GDM. Few women had a diagnosis of hypertension (2.2%), cardiovascular disease (5.0%), however, more women had a diagnosis of a mental health disorder (12.3%), or substance use disorder (9.2%) in the year prior to the birth of their child. The majority of women lived in counties that were classified as rural (54.0%), low income (59.1%), and as having a low level of education among AI/AN residents (72.5%). About half of the women resided in a county with low access to grocery stores (50.8%).

As shown in Table 1, 6.4% of infants in the study sample had macrosomia. The prevalence of macrosomia increased along with pre-pregnancy BMI category, such that macrosomia was more common for women with higher BMI. Macrosomia was also more common in post-term deliveries. No county-level differences in SDOH were noticeable between infant birth weight categories.

In Table 2, we present the adjusted odds ratios (OR) from multivariate generalized linear mixed models designed to evaluate the effects of maternal health factors and county-level SDOH characteristics on macrosomia. In Model 1, pre-pregnancy DM was significantly associated with higher odds of macrosomia (OR = 7.67; CI = 3.37,17.43, p < 0.0001). Meanwhile, compared to underweight/normal pre-pregnancy BMI, the odds of having an infant with macrosomia was significantly elevated among women with pre-pregnancy BMI categorized as overweight (OR = 3.09; CI = 1.17,8.13, p = 0.023) or obese (OR = 3.52; CI = 1.39,8.93, p = 0.008). In Model 2, in which we adjusted for major comorbidities, the odds of having an infant with high birth weight was higher when a woman had pre-pregnancy DM (OR = 8.33; CI = 3.56,19.47, p < 0.0001) or a pre-pregnancy BMI that was in the overweight (OR = 3.05; CI = 1.15,8.04, p = 0.025) or obese (OR = 3.52; CI = 1.38,8.95, p = 0.008) categories. In Model 3, when neighborhood and built environment measures were added, pre-pregnancy DM and pre-pregnancy overweight or obesity still increased the odds of having an infant with macrosomia. However, neighborhood and built environment factors were not associated with high infant birth weight.

In Model 4, in which county-level education and economic stability variables were added, we observed the largest odds ratio for having an infant with macrosomia with prepregnancy DM (OR = 10.31; CI = 4.19,25.37, p < 0.0001). Overweight (OR = 2.81, CI = 1.05, 7.51, p = 0.040) and obesity (OR = 3.43; CI = 1.33, 8.87, p = 0.011) were also associated with having increased odds of macrosomia. Further, rurality was associated with higher odds of macrosomia (OR = 4.01; CI = 1.02,15.70, p = 0.047). Finally, when the delivery was pre-term, the odds of having a high birthweight infant was significantly lower (OR = 0.13; CI = 0.02,0.97, p = 0.047).

Fig. 1 Flow chart defining study population and exclusions



	All	Infant low/ normal birth weight < 4,000 g n (%)	Infant high birth weight≥4000 g n (%)
Au	1,136 (100.0%)	1,063 (93.6%)	73 (6.4%)
Individual & healthcare			
Age group			
<25 years	552 (48.6%)	522 (94.6%)	30 (5.4%)
25-34 years	479 (42.2%)	445 (93.5%)	34 (7.1%)
35+years	105(9.2%)	229 (95.0%)	9 (8.6%)
Health insurance coverage			
Had no insurance coverage	182 (16.0%)	171 (94.0%)	11 (6.0%)
Had Medicaid	871 (76.7%)	814 (93.5%)	57 (6.5%)
Had private insurance	241 (21.2%)	229 (95.0%)	12 (5.0%)
Health condition			
Non-diabetic	945 (83.2%)	895 (94.7%)	50 (5.3%)
Prepregnancy diabetes <sup>a</sup>	46 (4.0%)	33 (71.7%)	13 (28.3%)
Gestational diabetes	145 (12.8%)	135 (93.1%)	10 (6.9%)
Prepregnancy BMI <sup>a</sup>			
Underweight or normal weight	243 (21.4%)	237 (97.5%)	6 (2.5%)
Overweight	205 (18.0%)	189 (91.3%)	16 (7.8%)
Obese	271 (23.9%)	244 (87.3%)	27 (10.0%)
No BMI measure	417 (36.7%)	393 (89.3%)	24 (5.8%)
Hypertension <sup>a</sup>	25 (2.2%)	22 (88.0%)	3 (12.0%)
Cardiovascular disease <sup>a</sup>	57 (5.0%)	54 (94.7%)	3 (5.3%)
Mental health disorder <sup>a</sup>	140 (12.3%)	133 (95.0%)	7 (5.0%)
Substance use disorder <sup>a</sup>	105 (9.2%)	99 (94.3%)	6 (5.7%)
Birth outcomes			
Pre-term delivery	72 (6.3%)	71 (98.1%)	1 (1.4%)
Post-term delivery	94 (8.3%)	86 (79.5%)	8 (8.5%)
Neighborhood & built environment			
Low access to grocery stores			
Counties with less access to the grocery store (% low access>25.3%)	577 (50.8%)	535 (92.7%)	42 (7.3%)
SNAP authorized stores per 1,000 population			
Counties above the median (0.8%)	503 (44.3%)	471 (93.6%)	32 (6.4%)
NCHS County rurality 2013	613 (54.6%)	566 (92.3%)	47 (7.7%)
Education & economic stability			
AI/AN Education: % adults < high school			
Counties with lower levels of educational attainment (% less than high school > 46.0%)	824 (72.5%)	766 (93.0%)	58 (7.0%)
AI/AN Income: % households < 139% FPL			
Counties with higher levels of poverty (% below 139% FPL>42.5%)	671 (59.1%)	626 (93.3%)	45 (6.7%)

 Table 1 Maternal descriptive characteristics stratified by infant birth weight

Abbreviations: AI/AN: American Indian/Alaska Native; BMI: Body mass index; FPL: Federal Poverty Level; g: grams; SNAP: Supplemental Nutrition Assistance Program; NCHS: National Center for Health Statistics <sup>a</sup> Data from FY2010 **Table 2** Adjusted Odds Ratios (ORs) from Multivariate Generalized Linear Mixed Models

 Evaluating the Effects of Maternal Health Factors and County-Level Social Determinates of

 Health

	Model 1	Model 2 OR (95% CI)	Model 3 OR (95% CI)	Model 4 OR (95% CI)
	OR (95% CI)			
Age group	10000	1.5.77.77.7	Advantage .	and the second
<25 years	1.01 (0.59,1.73)	1.02 (0.59,1.76)	0.73 (0.49,1.08)	0.94 (0.54,1.66)
25-34 years (reference)				
35+years	0.73 (0.31,1.72)	0.70 (0.29,1.69)	0.96 (0.53,1.74)	0.83 (0.33,2.09)
Health insurance coverage				
Medicaid	0.88 (0.48,1.62)	0.89 (0.48,1.63)	0.83 (0.55,1.25)	1.08 (0.57,2.04)
Private insurance	0.71 (0.37,1.36)	0.71 (0.37,1.37)	0.56 (0.32,0.95)	0.84 (0.41,1.72)
Health condition				
Prepregnancy diabetes	7.67 (3.37,17.43)***	8.33 (3.56,19.47)***	4.50 (2.14,9.45)***	10.31 (4.19,25.37)***
Gestational diabetes	1.16 (0.57,2.39)	1.18 (0.58,2.42)	0.93 (0.56,1.56)	1.28 (0.61,2.68)
Prepregnancy BMI				
Underweight or normal weight (reference)				
Overweight	3.09 (1.17,8.13)*	3.05 (1.15,8.04)*	2.79 (1.04,7.44)*	2.81 (1.05,7.51)*
Obese	3.52 (1.39,8.93)**	3.52 (1.38,8.95)**	3.40 (1.32,8.78)*	3.43 (1.33,8.87)*
No BMI measure	2.10 (0.84,5.27)	2.08 (0.83,5.21)	2.12 (0.82,5.46)	2.18 (0.83,4.30)
Hypertension		0.88 (0.28,2.77)	1.32 (0.32,5.41)	1.26 (0.31,5.23)
Cardiovascular disease		0.56 (0.20,1.62)	0.89 (0.26,3.13)	0.89 (0.25,3.12)
Mental health disorder		0.81 (0.48,1.37)	0.69 (0.29,1.63)	0.69 (0.29,1.65)
Substance use disorder		1.03 (0.60,1.75)	0.76 (0.29,1.98)	0.76 (0.29,1.99)
Birth outcomes				
Pre-term delivery	0.14 (0.02,1.05)	0.13 (0.02,1.02)	0.13 (0.02,0.97)	0.13 (0.02,0.97)*
Post-term delivery	1.57 (0.72,3.44)	1.54 (0.70,3.37)	1.91 (0.84,4.32)	1.89 (0.83,4.30)
Neighborhood & built environment				
Low access to grocery stores				
Counties with less access to the grocery store (% low access > 25.3%)			1.10 (0.76,3.36)	2.07 (0.37,11.66)
SNAP authorized stores per 1,000 pop				
Counties above the median (0.8%)			1.11 (0.38,3.25)	1.07 (0.35,3.25)
NCHS rural counties			2.55 (0.91,7.16)	4.01 (1.02.15.70)*
Education & economic stability				
AI/AN Education: % adults < high school				
Counties with lower levels of educational attainment (% less than high school>46.0%)				0.62 (0.17,2.31)
AI/AN Income: % households < 139% FPL				
Counties with higher levels of poverty (% below 139% FPL>42.5%)				0.47 (0.08.2.76)

Abbreviations: AI/AN: American Indian/Alaska Native; BMI: Body mass index; FPL: Federal Poverty Level; SNAP: Supplemental Nutrition Assistance Program; NCHS: National Center for Health Statistics

\*\*\* p < 0.001; \*\* p < 0.01; \* p < 0.05

#### Discussion

This study examined individual- and geographic-level factors associated with macrosomia in the AI/AN population, utilizing IHS electronic medical records and county-level measures of SDOH. We found a strong association of obesogenic influences present before and during gestation with macrosomia in our sample of AI/AN infants. Macrosomia was associated with maternal pre-pregnancy DM and prepregnancy overweight and obesity. GDM was not associated with increased odds of macrosomia in our study. Rural residence also increased odds of high infant birth weight.

Our results confirm findings from the few other studies examining pre-pregnancy DM, pre-pregnancy BMI, and infant birth outcomes in AI/AN populations. Dyck et al., (2019) conducted a retrospective, longitudinal, intra-intergenerational study among First Nations women and found that women with GDM and pre-GDM were more likely to have an infant with macrosomia (Dyck et al., 2019). Using national surveillance data from the Pregnancy Risk Assessment Monitoring System from AI/AN respondents with singleton births, Rockhill et al., (2015) found pre-pregnancy BMI and gestational weight gain were independent predictors of macrosomia among AI/AN women, with the prevalence of macrosomia in their sample at 14% (Rockhill et al., 2015). Anderson, Spicer, and Peercy (2017) used a population-based retrospective sample of live singleton births in the United States to examine the relationship between obesity, diabetes, and birth outcomes among AI/AN and other racial/ethnic groups and found overweight, obesity, and DM were all associated with increased odds of macrosomia (Anderson et al., 2016). We too found higher than normal BMI and DM prior to pregnancy to be associated with increased odds of high birthweight. However, in our sample, the odds ratios were much higher than odds ratios reported by either Rockhill et al. (adjusted OR = 1.27; CI =1.01,1.59 for macrosomia with pre-pregnancy overweight; adjusted OR = 1.63; CI = 1.29,2.07for macrosomia with pre-pregnancy obesity) or Anderson et al. (OR = 1.47; CI = 1.35, 1.61 for macrosomia with pre-pregnancy overweight; OR = 1.99; CI = 1.83, 2.16 for macrosomia with pre-pregnancy obesity). Furthermore, unlike the studies conducted by Dyck et al. and Anderson et al., we did not find an association of macrosomia with GDM.

Many early life risk factors for childhood and adult obesity are more prevalent among United States minority populations than among non-Hispanic whites and may explain the higher prevalence of obesity among racial/ethnic minority children. Pediatric obesity currently has a prevalence of 17% in the United States and, once present, is difficult to reverse (Ogden et al., 2014). National estimates of infant obesity range from 8 to 16%, with evidence that children who are obese at six months of age are likely to remain obese at 24 months (Ogden et al., 2014). Obesity in early childhood tends to persist through childhood and across the lifespan, resulting in increased individual, social, health, and economic costs (Han et al., 2010; Wang et al., 2008) determined that obesity will account for more than 16% of all US healthcare expenditures by 2030 (Wang et al., 2008), which is particularly concerning for the chronically underfunded IHS (Sequist et al., 2011). Our findings are important for chronic disease prevention in AI/AN populations, as prevention is universally viewed as the best approach for reversing the rising global prevalence of obesity. To date, limited evidence exists regarding the most effective means of preventing obesity in AI/AN children (Bahia et al., 2019; Wang et al., 2015). Identifying crucial periods of developmental programming is important for designing effective interventions, considering that overweight, obesity, and GDM are transgenerational health burdens in the AI/AN population (Bullock et al., 2017). Pre-pregnancy overweight and obesity and prepregnancy DM affected a substantial number of pregnancies in our sample and were associated with infant macrosomia. AI/AN culturally grounded interventions such as the Special Diabetes Program for Indians Diabetes Prevention (SDPI) demonstration project (Jiang et al., 2013), the Obesity Prevention and Evaluation of InterVention Effectiveness in NaTive North Americans (OPREVENT) multilevel multicomponent household intervention (Jock et al., 2022; Redmond et al., 2019), and the Tribal Health and Resilience in Vulnerable Environments (THRIVE) intervention (Love et al., 2019) have shown promising approaches to obesity prevention in AI/AN communities. Preconception counseling, or education of all women and adolescents of reproductive age, about the risks of entering pregnancy with risk factors, such as DM or obesity, and how pregnancy planning can improve both maternal and fetal outcomes is a standard of care recommended by the American Diabetes Association (American Diabetes Association, 2020). Providing preconception counseling and encouraging women to adopt a healthy lifestyle and maintain a healthy weight prior to pregnancy can reduce risk for GDM and poor pediatric outcomes (Moore et al., 2019; Nadeau et al., 2020). Postpartum interventions are also warranted to support AI/AN women (Jones et al., 2015).

Given the potential for a large impact on the population, future AI/AN overweight and obesity intervention trials aiming to reduce the prevalence of childhood and adult overweight and obesity should target pre-pregnancy weight status and diabetes prevention. Tribal nations, counties, states and the IHS need to address AI/AN overweight and obesity through multifactorial and comprehensive approaches that reckon with health inequities, colonial history, and funding inadequacies of policies and programs (Christensen & Damon, 2022; Dagher & Linares, 2022; Jaramillo et al., 2022).

In our study, most county-level SDOH variables, including AI/AN educational attainment, AI/AN household income, access to grocery stores, and SNAP authorized stores per capita, were not associated with high infant birth weight. The only county-level SDOH that was associated with macrosomia was county rurality. The associated between macrosomia and county rurality may be due in part to multiple co-occurring characteristics of the rural counties included in our analysis. The mechanism for this relationship is unclear but may be associated with SDOH and access, availability, and affordability of healthy foods in grocery stores. Jiang et al., (2018) found less BMI reduction and physical activity increase among AI/AN participants in the SDPI who lived in neighborhoods with higher concentrations of AI/ANs, when controlling for individual level socioeconomic status. Meta-analyses have determined that environmental characteristics including rurality and low access to grocery stores were consistently associated with obesity (Leal & Chaix, 2011). Many IHS beneficiaries reside on a reservation or near a reservation, and chose to live near ancestral lands with family and cultural ties (Jiang et al., 2018), areas that tend to be rural.

We did not find an association between high birthweight and GDM, despite an increased risk for macrosomia characteristic for GDM. Although insulin is the preferred medication for treating hyperglycemia in women with GDM, some women with GDM may be placed on metformin as an alternative therapy. Our study design did not collect data on treatment for GDM (Bahia et al., 2019). However, a 2019 systemic review and meta-analysis of the impact of metformin in GDM upon the offspring growth trajectory showed that infants born to women with GDM treated with metformin weighed 108 gm lower on average than infants born to women with GDM treated with insulin (Tarry-Adkins et al., 2019). Some high-risk women in our sample with preexisting DM or GDM may have been referred elsewhere for prenatal care and/or delivery, and thus have been excluded from our sample.

This work had strengths and limitations. A notable strength was our ability to assess both individual- and county-level variables related to infant birthweight. This study includes data on dyads who were eligible for and received health care at IHS facilities (i.e., are tribally enrolled and who elected to receive care at IHS facilities) located in six areas. Limitations included not being able to successfully link women and newborns at four of the 10 sites identified for study inclusion. Thus, the findings from our study may not be generalizable to AI/AN women and infants who receive services from other IHS Service Units or who are not eligible or access IHS services (Indian Health Service, 2010; US Census Bureau, 2015, 2016). While studies have documented future health risks associated with low birthweight (Feng et al., 2018; Knop et al., 2018), we did not investigate low birthweight, as our sample had a limited number of infants with low birthweight (N = 42). Finally, individual-level education, income, family health history, and gender identity variables were not available and hence we were not able to control for individual level socioeconomic status in our models.

## Conclusion

Previous studies within AI/AN populations on macrosomia have not included social determinants of health risk factors. The current study found that AI/AN women who were overweight or obese, had pre-pregnancy diabetes, or lived in a rural county had higher odds of having an infant with a high birthweight. Our findings have implications for interventions to disrupt the cycle of obesity and adverse metabolic changes and their sequelae in subsequent generations of AI/AN people. Multigenerational interventions targeting prepregnancy weight reduction and control in a culturally appropriate manner are needed to support healthy weight. The role of rurality as a characteristic of the setting in which AI/AN people live and interventions take place should be further explored and considered in intervention planning and evaluation.

Acknowledgements Data used in this study are from the Indian Health Service (IHS) Health Care Delivery Data Project; the data include information from many American Indian and Alaska Native communities. Data Project analyses are conducted with the guidance and advice of IHS and Tribal health program colleagues, as well as members of the project's Steering, Project Site, and Patient Committees. Members of Tribal and IHS institutional review boards, Tribal Councils, and Tribal Authorities educate us about the health concerns they have for their Tribal members and how they hope this project will inform their work. This project relies on their support and approval. The authors would also like to express their gratitude to Sara Mumby for her editorial assistance.

**Funding Statement** The research reported in this publication was supported by the National Institute of Diabetes and Digestive and Kidney Diseases (NIDDK) of the National Institutes of Health under Award Numbers P30DK092923 (S.M. Manson) and R18DK114757 (J.M. O'Connell and S.M. Manson). Funding for the development of the data infrastructure, utilized in the reported analyses, was supported by the Agency for Healthcare Research and Quality (290-2006-00020-I, TO #11, J.M. O'Connell) and the Patient-Centered Outcomes Research Institute (AD-1304-6451, J.M. O'Connell). The content of this report is solely the responsibility of the authors and does not necessarily represent the official views of these organizations.

# References

- American College of Obstetricians and Gynecologists, Macrosomia, O., & Gynecology : January 2020 Volume 135 Issue 1 pe18-e35doi: https://doi.org/10.1097/AOG.00000000003606
- American Diabetes Association. (2020). Management of Diabetes in Pregnancy: Standards of Medical Care in Diabetes-2020. Diabetes Care, 43(Suppl 1), S183–s192. https://doi.org/10.2337/dc20-S014
- Anderson, K. G., Spicer, P., & Peercy, M. T. (2016). Obesity, Diabetes, and Birth Outcomes Among American Indians and Alaska Natives. Maternal Child Health Journal, 20(12), 2548–2556. https://doi.org/10.1007/s10995-016-2080-3
- Bahia, L., Schaan, C. W., Sparrenberger, K., Abreu, G. A., Barufaldi, L. A., Coutinho, W., & Schaan, B. D. (2019). Overview of meta-analysis on prevention and treatment of childhood obesity. Jornal de Pediatria, 95(4), 385–400. https://doi.org/10.1016/j.jped.2018.07.009
- Boney, C. M., Verma, A., Tucker, R., & Vohr, B. R. (2005). Metabolic syndrome in childhood: association with birth weight, maternal obesity, and gestational diabetes mellitus. Pediatrics, 115(3), e290–296. https://doi.org/10.1542/peds.2004-1808
- Brenneman, G., Rhoades, E., & Chilton, L. (2006). Forty years in partnership: the American Academy of Pediatrics and the Indian Health Service. Pediatrics, 118(4), e1257–1263. https://doi.org/10.1542/peds.2006-0362
- Bullock, A., Sheff, K., Moore, K., & Manson, S. (2017). Obesity and Overweight in American Indian and Alaska Native Children, 2006–2015. American Journal of Public Health, 107(9), 1502–1507. https://doi.org/10.2105/AJPH.2017.303904
- Carter, M. A., Dubois, L., Tremblay, M. S., Taljaard, M., & Jones, B. L. (2012). Trajectories of childhood weight gain: the relative importance of local environment versus individual social and early life factors. PLoS One, 7(10), e47065. https://doi.org/10.1371/journal.pone.0047065
- Chang, M. H., Moonesinghe, R., Athar, H. M., & Truman, B. I. (2016). Trends in Disparity by Sex and Race/Ethnicity for the Leading Causes of Death in the United States-1999-2010. Journal of Public Health Management and Practice, 22(Suppl 1), S13–24. https://doi.org/10.1097/PHH.00000000000267
- Christensen, L., & Damon, S. (2022). Social Determinants of Health and Response to Disease Associated With Health Outcomes of American Indian and Alaska Native Patients. JAMA Network Open, 5(3), e224827. https://doi.org/10.1001/jamanetworkopen.2022.4827
- Dagher, R. K., & Linares, D. E. (2022). A Critical Review on the Complex Interplay between Social Determinants of Health and Maternal and Infant Mortality. Children 9(3). https://doi.org/10.3390/children9030394
- Dixon, B., Pena, M. M., & Taveras, E. M. (2012). Lifecourse approach to racial/ethnic disparities in childhood obesity. Advances in Nutrition, 3(1), 73–82. https://doi.org/10.3945/an.111.000919
- Duryea, E. L., Hawkins, J. S., McIntire, D. D., Casey, B. M., & Leveno, K. J. (2014). A revised birth weight reference for the United States. Obstetrics & Gynecology, 124(1), 16–22. https://doi.org/10.1097/AOG.0000000000345
- Dyck, R. F., Karunanayake, C., Pahwa, P., Stang, M., & Osgood, N. D. (2019). Epidemiology of

Diabetes in Pregnancy Among First Nations and Non-First Nations Women in Saskatchewan, 19802013. Part 2: Predictors and Early Complications; Results From the Diabetes in Pregnancy: Outcomes and Ramification from a Retrospective Longitudinal Intra- Intergenerational Study in Saskatchewan Project. Canadian Journal of Diabetes, 44(7), 605–614. https://doi.org/10.1016/j.jcjd.2019.11.001

- Feng, C., Osgood, N. D., & Dyck, R. F. (2018). Low Birth Weight Cumulative Obesity Dose, and the Risk of Incident Type 2 Diabetes. Journal of Diabetes Research, 2018, 8435762. https://doi.org/10.1155/2018/8435762
- Garcia-Vargas, L., Addison, S. S., Nistala, R., Kurukulasuriya, D., & Sowers, J. R. (2012). Gestational Diabetes and the Offspring: Implications in the Development of the Cardiorenal Metabolic Syndrome in Offspring. Cardiorenal Med, 2(2), 134–142. https://doi.org/10.1159/000337734
- Halvorson, K. L., Vogt, H. B., Kightlinger, L., & Stevens, D. (2017). The Impact of Maternal Diabetes, Obesity and Race on Infant Birth Weights in South Dakota. South Dakota Journal of Medicine, 70(2), 61–66. https://www.ncbi.nlm.nih.gov/pubmed/28810088
- Han, J. C., Lawlor, D. A., & Kimm, S. Y. (2010). Childhood obesity. Lancet, 375(9727), 1737– 1748. https://doi.org/10.1016/S0140-6736(10)60171-7
- Hu, C., Mu, Y., Wan, Q., Qin, Y., Hu, R., Shi, L., Su, Q., Yu, X., Yan, L., Qin, G., Tang, X., Chen, G., Gao, Z., Wang, G., Shen, F., Luo, Z., Chen, L., Huo, Y., Li, Q., & Group, R. S. (2020). Association between birth weight and diabetes: Role of body mass index and lifestyle in later life. Journal of Diabetes, 12(1), 10–20. https://doi.org/10.1111/1753-0407.12960
- Hunt, K. J., Alanis, M. C., Johnson, E. R., Mayorga, M. E., & Korte, J. E. (2013). Maternal prepregnancy weight and gestational weight gain and their association with birthweight with a focus on racial differences. Maternal Child Health Journal, 17(1), 85–94. https://doi.org/10.1007/s10995-012-0950-x
- Indian Health Service (2010). User Population Report (HQ Summary), Three Year; Age Groups and Sex by Areas and Service Units of Residence, Fiscal Year 2010.
- Jaramillo, E. T., Haozous, E., & Willging, C. E. (2022). The Community as the Unit of Healing: Conceptualizing Social Determinants of Health and Well-Being for Older American Indian Adults. The Gerontologist, 62(5), 732–741. https://doi.org/10.1093/geront/gnac018
- Jiang, L., Chang, J., Beals, J., Bullock, A., & Manson, S. M. (2018). Neighborhood characteristics and lifestyle intervention outcomes: Results from the Special Diabetes Program for Indians. Preventative Medicine, 111, 216–224. https://doi.org/10.1016/j.ypmed.2018.03.009
- Jiang, L., Manson, S. M., Beals, J., Henderson, W. G., Huang, H., Acton, K. J., & Roubideaux, Y. (2013). Translating the Diabetes Prevention Program into American Indian and Alaska Native communities: results from the Special Diabetes Program for Indians Diabetes Prevention demonstration project. Diabetes Care, 36(7), 2027–2034. https://doi.org/10.2337/dc12-1250
- Jock, B. W. I., Maudrie, T., Fleischhacker, S., Porter, K. P., & Gittelsohn, J. (2022). Journey to Promoting Structural Change for Chronic Disease Prevention: Examining the Processes for Developing Policy, Systems, and Environmental Supports in Native American Nations. Current Developments in Nutrition, 6(3), nzab031. https://doi.org/10.1093/cdn/nzab031

- Jones, D. S. (2006). The persistence of American Indian health disparities. American Journal of Public Health, 96(12), 2122–2134. https://doi.org/10.2105/ajph.2004.054262
- Jones, E. J., Peercy, M., Woods, J. C., Parker, S. P., Jackson, T., Mata, S. A., McCage, S., Levkoff, S. E., Nicklas, J. M., & Seely, E. W. (2015). Identifying postpartum intervention approaches to reduce cardiometabolic risk among American Indian women with prior gestational diabetes, Oklahoma, 2012–2013. Preventing Chronic Disease, 12, E45. https://doi.org/10.5888/pcd12.140566
- Knop, M. R., Geng, T. T., Gorny, A. W., Ding, R., Li, C., Ley, S. H., & Huang, T. (2018). Birth Weight and Risk of Type 2 Diabetes Mellitus, Cardiovascular Disease, and Hypertension in Adults: A Meta-Analysis of 7 646 267 Participants From 135 Studies. Journal of the American Heart Association, 7(23), e008870. https://doi.org/10.1161/jaha.118.008870
- Kruse, G., Lopez-Carmen, V. A., Jensen, A., Hardie, L., & Sequist, T. D. (2022). The Indian Health Service and American Indian/Alaska Native Health Outcomes. Annual Reviews of Public Health, 43, 559–576. https://doi.org/10.1146/annurev-publhealth-052620-103633
- Lanier, A. P. (2007). Existence of Alaska Native health disparities. American Journal of Public Health, 97(9), 1541–1542. https://doi.org/10.2105/ajph.2007.115006. author reply 1542–1543
- LaVallie, D. L., Gabbe, S. G., Grossman, D. C., Larson, E. B., Baldwin, L. M., & Andrilla, C. H. (2003). Birth outcomes among American Indian/Alaska Native women with diabetes in pregnancy. Journal of Reproductive Medicine, 48(8), 610–616. https://www.ncbi.nlm.nih.gov/pubmed/12971141
- Leal, C., & Chaix, B. (2011). The influence of geographic life environments on cardiometabolic risk factors: a systematic review, a methodological assessment and a research agenda. Obesity Reviews, 12(3), 217–230. https://doi.org/10.1111/j.1467-789X.2010.00726.x
- Link, B. G., & Phelan, J. (1995). Social conditions as fundamental causes of disease. Journal of Health and Social Behavior, Spec No,80–94. https://www.ncbi.nlm.nih.gov/pubmed/7560851
- Love, C. V., Taniguchi, T. E., Williams, M. B., Noonan, C. J., Wetherill, M. S., Salvatore, A. L., Jacob, T., Cannady, T. K., Standridge, J., Spiegel, J., & Jernigan, V. B. B. (2019). Diabetes and Obesity Associated with Poor Food Environments in American Indian Communities: the Tribal Health and Resilience in Vulnerable Environments (THRIVE) Study. Current Developments in Nutrition, 3(Suppl 2), 63–68. https://doi.org/10.1093/cdn/nzy099
- Lynch, J., & Smith, G. D. (2005). A life course approach to chronic disease epidemiology. Annual Reviews of Public Health, 26, 1–35. https://doi.org/10.1146/annurev.publhealth.26.021304.144505
- Marmot, M., Friel, S., Bell, R., Houweling, T. A., & Taylor, S. (2008). Closing the gap in a generation: health equity through action on the social determinants of health. Lancet, 372(9650), 1661–1669. https://doi.org/10.1016/s0140-6736(08)61690-6
- Marshall, S. M. (2019). A life course perspective on diabetes: developmental origins and beyond. Diabetologia, 62(10), 1737–1739. https://doi.org/10.1007/s00125-019-4954-6
- Mitchell, F. M. (2012). Reframing diabetes in American Indian communities: a social determinants of health perspective. Health & Social Work, 37(2), 71–79. https://doi.org/10.1093/hsw/hls013
- Moore, K., Stotz, S., Fischl, A., Beirne, S., McNealy, K., Abujaradeh, H., & Charron-

Prochownik, D. (2019). Pregnancy and Gestational Diabetes Mellitus (GDM) in North American Indian Adolescents and Young Adults (AYA): Implications for Girls and Stopping GDM. Current Diabetes Reports, 19(11), 113. https://doi.org/10.1007/s11892-019-1241-3

- Nadeau, K. J., Stotz, S. A., Moore, K., Garcia-Reyes, Y., Sereika, S. M., Stein, H., & Charron-Prochownik, D. (2020). Beta Testing of a Gestational Diabetes Risk Reduction Intervention for American Indian and Alaska Native Teens. Journal of Pediatric Health Care, 34(5), 418–423. https://doi.org/10.1016/j.pedhc.2020.04.005
- O'Connell, J., Guh, S., Ouellet, J., Li, Y., Rockell, J., Croy, C., & Gutilla, M. (2014). ARRA ACTION: Comparative Effectiveness of Health Care Delivery Systems for American Indians and Alaska Natives Using Enhanced Data Infrastructure: Final Report. http://www.ahrq.gov/professionals/systems/system/deliverysysteminitiative/ihs/index.html
- Ogden, C. L., Carroll, M. D., Fryar, C. D., & Flegal, K. M. (2015). Prevalence of Obesity Among Adults and Youth: United States, 2011–2014.NCHS Data Brief(219),1–8. https://www.ncbi.nlm.nih.gov/pubmed/26633046
- Ogden, C. L., Carroll, M. D., Kit, B. K., & Flegal, K. M. (2014). Prevalence of childhood and adult obesity in the United States, 2011–2012. Jama, 311(8), 806–814. https://doi.org/10.1001/jama.2014.732
- Pachucki, M. C., & Goodman, E. (2015). Social Relationships and Obesity: Benefits of Incorporating a Lifecourse Perspective. Current Obesity Reports, 4(2), 217–223. https://doi.org/10.1007/s13679-015-0145-z
- Redmond, L. C., Jock, B., Gadhoke, P., Chiu, D. T., Christiansen, K., Pardilla, M., Swartz, J., Platero, H., Caulfield, L. E., & Gittelsohn, J. (2019). OPREVENT (Obesity Prevention and Evaluation of InterVention Effectiveness in NaTive North Americans): Design of a Multilevel, Multicomponent Obesity Intervention for Native American Adults and Households. Current Developments in Nutrition, 3(Suppl 2), 81–93. https://doi.org/10.1093/cdn/nz2009
- Rockhill, K., Dorfman, H., Srinath, M., & Hogue, C. (2015). The Effects of Prepregnancy Body Mass Index and Gestational Weight Gain on Fetal Macrosomia Among American Indian/ Alaska Native Women. Maternal Child Health Journal, 19(11), 2480–2491. https://doi.org/10.1007/s10995-015-1769-z
- Rosa, F., & Resnick, L. (1965). Birth Weight and Perinatal Mortality in the American Indian. American Journal of Obstetrics and Gynecology, 91, 972–976. https://doi.org/10.1016/0002-9378(65)90564-8

Institute, S. A. S. (2013). I. SAS software In (Version 9.4)

- Schell, L. M., & Gallo, M. V. (2012). Overweight and obesity among North American Indian infants, children, and youth. American Journal of Human Biology, 24(3), 302–313. https://doi.org/10.1002/ajhb.22257
- Sequist, T. D., Cullen, T., & Acton, K. J. (2011). Indian health service innovations have helped reduce health disparities affecting american Indian and alaska native people. Health Affairs, 30(10), 1965–1973. https://doi.org/10.1377/hlthaff.2011.0630
- Service, U. D., & o., A. E. R. (2010). Food Environment Atlas USDA Retrieved Accessed June 6, 2018 from www.ers.usda.gov/dataproducts/food-environment-atlas.aspx

- Shi, Y., De Groh, M., & Morrison, H. (2013). Perinatal and early childhood factors for overweight and obesity in young Canadian children. Canadian Journal of Public Health, 104(1), e69–74. https://doi.org/10.1007/bf03405658
- StataCorp (2015). Stata Statistical Software: Release 14. In StataCorp LP
- Statistics, N. C. H. (2017). 6/1/2017). NCHS Urban-Rural Classification Scheme for Counties. Retrieved 9/3/2020 from https://www.cdc.gov/nchs/data\_access/urban\_rural.htm
- Tarry-Adkins, J. L., Aiken, C. E., & Ozanne, S. E. (2019). Neonatal, infant, and childhood growth following metformin versus insulin treatment for gestational diabetes: A systematic review and meta-analysis. PLoS Medicine, 16(8), e1002848. https://doi.org/10.1371/journal.pmed.1002848
- Thierry, J., Brenneman, G., Rhoades, E., & Chilton, L. (2009). History, law, and policy as a foundation for health care delivery for American Indian and Alaska Native children. Pediatric Clinics of North America, 56(6), 1539–1559. https://doi.org/10.1016/j.pcl.2009.09.018
- Tian, G., Guo, C., Li, Q., Liu, Y., Sun, X., Yin, Z., Li, H., Chen, X., Liu, X., Zhang, D., Cheng, C., Liu, L., Liu, F., Zhou, Q., Wang, C., Li, L., Wang, B., Zhao, Y., Liu, D., & Hu, D. (2019). Birth weight and risk of type 2 diabetes: A dose-response meta-analysis of cohort studies. Diabetes Metabolism Research and Reviews, 35(5), e3144. https://doi.org/10.1002/dmrr.3144
- US Census Bureau (2001). Census 2000 Summary File 1 & Summary File 2 United States. Retrieved Accessed June 10, 2019 from https://www.census.gov/main/www/cen2000.html
- US Census Bureau (2015). Census 2010 Summary File 1 & Summary File 2 United States. Retrieved Accessed June 10, 2019 from https://www.census.gov/population/www/cen2010/glance/
- US Census Bureau (2016). 2010–2014 American Community Survey—United States. Retrieved Special Tabulation Released on September 26, 2016 from https://www.census.gov/programs-surveys/acs/news/data-releases/2014/release.htm
- Vaiserman, A. M. (2017). Early-Life Nutritional Programming of Type 2 Diabetes: Experimental and Quasi-Experimental Evidence. Nutrients, 9(3), https://doi.org/10.3390/nu9030236
- Verisk Health, I. (2011). Sightlines<sup>TM</sup> DxCG Risk Solutions. In (Version Version 4.0.1.)
- Voerman, E., Santos, S., Golab, P., Amiano, B., Ballester, P., Barros, F., Bergström, H., Charles, A., Chatzi, M. A., Chevrier, L., Chrousos, C., Corpeleijn, G. P., Costet, E., Crozier, N., Devereux, S., Eggesbø, G., Ekström, M., Fantini, S., Farchi, M. P., & Jaddoe, S., V. W. V (2019). Maternal body mass index, gestational weight gain, and the risk of overweight and obesity across childhood: An individual participant data meta-analysis. PLoS Medicine, 16(2), e1002744. https://doi.org/10.1371/journal.pmed.1002744
- Wang, Y., Beydoun, M. A., Liang, L., Caballero, B., & Kumanyika, S. K. (2008). Will all Americans become overweight or obese? estimating the progression and cost of the US obesity epidemic. Obesity (Silver Spring, Md.), 16(10), 2323–2330. https://doi.org/10.1038/oby.2008.351
- Wang, Y., Cai, L., Wu, Y., Wilson, R. F., Weston, C., Fawole, O., Bleich, S. N., Cheskin, L. J., Showell, N. N., Lau, B. D., Chiu, D. T., Zhang, A., & Segal, J. (2015). What childhood obesity prevention programmes work? A systematic review and meta-analysis. Obesity Reviews, 16(7), 547–565. https://doi.org/10.1111/obr.12277

Yu, Z. B., Han, S. P., Zhu, G. Z., Zhu, C., Wang, X. J., Cao, X. G., & Guo, X. R. (2011). Birth weight and subsequent risk of obesity: a systematic review and meta-analysis. Obesity Reviews, 12(7), 525–542. https://doi.org/10.1111/j.1467-789X.2011.00867.x

**Publisher's Note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Springer Nature or its licensor holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.

## **Authors and Affiliations**

Vanessa Y. Hiratsuka<sup>1,2</sup>; Margaret Reid<sup>3</sup>; Jenny Chang<sup>4</sup>; Luohua Jiang<sup>5</sup>; Angela G. Brega<sup>6</sup>; Amber L. Fyfe-Johnson<sup>7</sup>; Kimberly R. Huyser<sup>8</sup>; Michelle Johnson-Jennings<sup>9</sup>; Cheryl Conway<sup>10</sup> ; John F. Steiner1<sup>1</sup>; Jennifer Rockell<sup>12</sup>; Denise A. Dillard<sup>1</sup>; Kelly Moore<sup>13</sup>; Spero M. Manson<sup>13</sup> ; Joan O'Connell<sup>14</sup>

Vanessa Y. Hiratsuka vyhiratsuka@alaska.edu

Margaret Reid margaret.reid@cuanschutz.edu

Jenny Chang jjchang@hs.uci.edu

Luohua Jiang lhjiang@hs.uci.edu

Angela G. Brega Angela.Brega@cuanschutz.edu

Amber L. Fyfe-Johnson afj@wsu.edu

Kimberly R. Huyser kimberly.huyser@ubc.ca

Michelle Johnson-Jennings mjohnsonjennings@usask.ca

Cheryl Conway cmconway92@gmail.com John F. Steiner john.f.steiner@kp.org

Jennifer Rockell jrockell@telligen.com

Denise A. Dillard dadillard@scf.cc

Kelly Moore Kelly.Moore@cuanschutz.edu

Spero M. Manson Spero.Manson@cuanschutz.edu

Joan O'Connell Joan.OConnell@CUAnschutz.edu

- 1 Research Department, Southcentral Foundation, 4501 Diplomacy Drive, 99508 Anchorage, AK, USA
- 2 Center for Human Development, University of Alaska Anchorage, 3211 Providence Drive, 99508, 99504 Anchorage, AK, USA
- 3 Department of Health Systems, Management, and Policy, Colorado School of Public Health, University of Colorado Denver, Denver, USA
- 4 Department of Medicine, School of Medicine, University of California, 301 Medical Surge II, 92697-7550 Irvine, CA, USA
- 5 Department of Epidemiology, University of California, Irvine, 3076 AIRB, 92697-7550 Irvine, CA, USA
- 6 School of Public Health, Centers for American Indian and Alaska Native Health, University of Colorado Anschutz Medical Campus, 13055 East 17th Avenue, 80045 Aurora, Colorado, CO, USA
- Institute for Research and Education to Advance Community Health (IREACH),
   Department of Medical Education and Clinical Sciences, Washington State
   University, 1100 Olive Way, Ste 1200, 98101 Seattle, WA, USA
- 8 Department of Sociology, The University of British Columbia, Vancouver, USA
- 9 Canada Research Chair for Indigenous Community Engaged Research for Indigenous Community Engaged Research, LE Clinical Health Psychologist, University of Saskatchewan, University of Colorado- Associate Professor, University of Washington- Associate Professor, Washington, USA
- 10 NE-BC; Quality Consultant, Charles George Veterans Medical Center, Asheville, NC, USA
- 11 Institute for Health Research, Department of Medicine, Kaiser Permanente Colorado, University of Colorado Anschutz Medical Campus, Aurora, USA
- 12 Telligen, Inc. Greenwood Village, 7730 E. Belleview Ave Suite 300 Greenwood, 80111 Village, CO, USA

- 13 Centers for American Indian and Alaska Native Health, University of Colorado, Anschutz Medical Campus, Mail Stop F800, 13055 E. 17th Avenue, 80045 Aurora, CO, USA
- Colorado School of Public Health, Centers for American Indian and Alaska Native Health, University of Colorado, Anschutz Medical Campus, Mail Stop F800, 13055
   E. 17<sup>th</sup> Avenue, 80045 Aurora, CO, USA